Author's response to reviews

Title: Integration of HIV care into maternal health services: a crucial change required in improving quality of obstetric care in high HIV prevalence countries

Authors:

   Farai D Madzimbamuto (faraitose@hotmail.com)
   Sunanda C Ray (sunanda28@hotmail.com)
   Keitshokile D Mogobe (mogobekd@mopipi.ub.bw)

Version: 3 Date: 14 March 2013

Author's response to reviews:

Response to reviewers in cover letter